Author's response to reviews

Title: Factors influencing the growth hormone peak and plasma insulin-like growth factor I in young adults with pituitary stalk interruption syndrome

Authors:

Mariana Marcu (marimarcu@yahoo.com)
Christine Trivin (christine.trivin@nck.aphp.fr)
Jean-Claude Souberbielle (jean-claude.souberbielle@nck.aphp.fr)
Raja Brauner (raja.brauner@wanadoo.fr)

Version: 3 Date: 15 May 2008

Author's response to reviews: see over
Factors influencing the growth hormone peak and plasma insulin-like growth factor I in young adults with pituitary stalk interruption syndrome

Version 3

To Dr Stefano Zucchi

Thank you for reviewing our paper.

Major Compulsory Revisions
The GHRH plus arginine test has been added to the “Analysis of the limitations of the study”.
The sentence summarizing the main findings of the study is now at the beginning of the Discussion.

Minor essential Revisions
The English language has been reviewed by Dr Owen Parkes.

Discretionary Revisions
We do not wish to change the title.